Author's response to reviews

Title: Geographical information system and predictive risk maps of urinary schistosomiasis in Ogun State, Nigeria

Authors:

Uwem F Ekpo (ufekpo@hotmail.com)
Chiedu F Mafiana (chiedu.mafiana@gmail.com)
Clement O Adeofun (clemluv2000@yahoo.com)
Adewale RT Solarin (solarinart@unaab.edu.ng)
Adewumi B Idozu (idowuab@unaab.edu.ng)

Version: 4 Date: 24 May 2008

Author's response to reviews: see over
24 May 2008

Editor-in-chief

BMC Infectious Diseases

Dear Editor,

We are pleased to know that our revised manuscript has been accepted for publication in principle. We once again thank the reviewers for their useful comments. The following formatting changes have been effected as required by the Journal.

1. Introduction has been renamed ‘Background’.
2. Page 4: FAO been changed to ‘Food and Agricultural Organization (FAO)’
3. Page 6: Section heading Population Data changed to ‘Population data’
4. Page 8, last line: GIS changed to ‘ArcView GIS 3.2a’
5. Page 12, line 17: state changed to ‘Ogun State Government’
6. Page 13, line 1: Abeokuta North LGA changed to ‘Abeokuta North LGA, Ogun State’
7. Page 13, line 7: first step reactivating change to ‘first step in reactivating’
8. Page 14: Acknowledgement: part of UFE doctoral project changed to ‘doctoral project of UFE’
9. Page 20: Title of Table 1 changed to ‘Prevalence of reported urinary schistosomiasis by LGA in Ogun State, Nigeria’

We await further directive from the editor

Thank you

Dr Uwem Ekpo

Corresponding Author